Author's response to reviews

Title: Immediate and One-Year Outcome of Patients Presenting with Acute Coronary Syndrome Complicated by Stroke: Findings from the 2nd Gulf Registry of Acute Coronary Events (Gulf RACE-2)

Authors:

- Jassim Al Suwaidi (jalsuwaidi@hotmail.com)
- Khalid Al Habib (khalidalhabib13@hotmail.com)
- Nidal Asaad (nidalasaad@gmail.com)
- Rajvir Singh (rajvir.aiims@gmail.com)
- Ahmad Hersi (alhersi@hotmail.com)
- Husam Al Falaeh (halfaleh@hotmail.com)
- Shukri Al Saif (shukrialsaif@yahoo.com)
- Ahmed Al-Motarreb (amotarreb2@gmail.com)
- Wael Almahmeed (walmahmeed@skmc.ae)
- Kadhim Sulaiman (kjsulaiman@gmail.com)
- Haitham Amin (haitham_amin@yahoo.com)
- Jawad Al-Lawati (jallawati@gmail.com)
- Norah Q. Al-Sagheer (nora_78@maktoob.com)
- Alawi A. Alsheikh-Ali (alsheikhali@gmail.com)

Version: 2 Date: 14 December 2011

Author's response to reviews: see over
December, 14, 2011

Lars Køber, MD
BMC Cardiovascular Disorders Journal
Section Editor

Dear Professor Køber

We would be honored if you could kindly accept submission of our manuscript titled “Immediate and One-Year Outcome of Patients Presenting with acute coronary syndrome complicated by stroke: Findings from the 2nd Gulf Registry of Acute Coronary Events (Gulf RACE-2) for consideration of publication in the BMC Cardiovascular Disorders Journal. This registry included consecutive patients with acute coronary syndrome in 6 Middle eastern countries involving 64 hospitals. In the revised manuscript, in the methods section we modified the ethical requirements as was suggested and we added author contribution section as was suggested.

Sincerely,

Jassim Al Suwaidi, MB,ChB,FACC,FESC,FSCAI
Associate Professor of Medicine
Weill Cornell Medical School
Department of Cardiology and Cardiovascular Surgery
Hamad Medical Corporation, Doha, Qatar